Author's response to reviews

Title: Hypothesis: primary antiangiogenic method proposed to treat early stage breast cancer

Authors:

   Michael W Retsky (michael.retsky@childrens.harvard.edu)
   William JM Hrushesky (william.hrushesky@med.va.gov)
   Isaac D Gukas (i.gukas@uea.ac.uk)

Version: 7 Date: 7 January 2009

Author's response to reviews:

figure was cropped to remove extra white space
references were put in required format
extra files were removed
author names were changed to reflect required format
font changed to Ariel 12 pt